

## Format for ANSWERING REVIEWERS

November 01, 2014



Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: theodorakis et al ET1 PHT WJG revised doc).

**Former Title:** Endothelin-1 hypo-response in murine portal hypertension is eNOS and iNOS independent

**New Title:** Murine study of portal hypertension associated endothelin-1 hypo-response

**Author:** Nicholas G. Theodorakis, Mary A. Maluccio and Nicholas J. Skill,

**Name of Journal:** *World Journal of Gastroenterology*

**ESPS Manuscript NO:** 13827

The manuscript has been improved according to the suggestions of reviewers:

**Revision to the text has been made according to the suggestions of the reviewer,**

Title was changed.

Introduction was edited as per recommend by reviewers

Author contribution added

Discussion edited

Illustrative figure added.

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

*Nicholas J Skill*

Nicholas Skill PhD  
Assistant Research Professor  
Indiana University  
C519 R3. 980 West Walnut Street  
Indianapolis, IN, 46077